Author's response to reviews

Title: Temporomandibular Joint Disc Repositioning Using Bone Anchors: An Immediate Post Surgical Evaluation by Magnetic Resonance Imaging

Authors:

ShanYong zhang (zhangshanyong@126.com)
XiuMing Liu (X.M.Liu@yahoo.com.cn)
XiuJuan Yang (X.J.Yang@126.com)
Chi Yang (yangchi63@hotmail.com)
MinJie Chen (M.J.Chen@126.com)
Majd S Haddad (M.S.haddad@uiowa.edu)
ZhuoZhi Chen (Z.Z.Chen@hotmail.com)

Version: 11 Date: 12 November 2010

Author's response to reviews: see over
Dear editor:

Thank you very much for acceptance my paper in your Journal again (1249876804304315). The paper was carefully formatted according to the editors’ suggestion in the proofing stage.

Best wishes,

Yours Sincerely

Associate Professor: ShanYong ZHANG, MD, DDS

Department of OMS

Ninth People’s Hospital

School of Medicine

Shanghai Jiaotong University

Zhi-Zao-Ju Rd., No. 639

Shanghai 200011, P.R. China

zhangshanyong@126.com
**Formatting changes to be made:**

I have made these changes:

Conclusions: add a Conclusions heading following the Discussion.

Abstract: I have changed the Purpose to Background.

Introduction: rename Background.

The figure 2 is cropped as closely as possible to minimise white space around the image.

And the figure 3 is changed into a new one

I have removed the additional files

And in addition, I have checked my manuscript for any typographical errors during the final proofing stage for my manuscript